Author’s response to reviews

Title: The application of wide-field laser ophthalmoscopy in fundus examination before myopic refractive surgery

Authors:

Lin Liu (sdiulin60@126.com)
Fang Wang (milwang_122@msn.com)
Ding Xu (daisyxu70@hotmail.com)
Chunlei Xie (xiechunlei86@126.com)
Jun Zou (zoujun70@126.com)

Version: 4 Date: 23 Nov 2017

Author’s response to reviews:

Point-by-point response to reviewers

Editor Comments:

The statement "All the data supporting our findings is contained within the manuscript" is invalid. This is because Availability of Data and Materials refers to the raw data-set upon which the summary statistics in the manuscript are based.

Answer: Thanks, following your advice, we have replaced “All the data supporting our findings is contained within the manuscript” by “The datasets analysed during the current study available from the corresponding author on reasonable request (zoujun70@126.com)” in revised manuscript (See in the Abbreviations).